Author’s response to reviews

Title: Prevalence of fibromyalgia in low socioeconomic status population

Authors:

Ana A. Assumpção (anassumpcao@usp.br)
Alane B Cavalcante (alanebc@gmail.com)
Cristina E Capela (criscapela@terra.com.br)
Juliana F Sauer (juliana.sauer@gmail.com)
Suellen D Chalot (ischalot@uol.com.br)
Carlos AB Pereira (cadebp@gmail.com)
Amélia P Marques (pasqual@usp.br)

Version: 3 Date: 15 November 2008

Author’s response to reviews: see over
São Paulo, Brazil, 1st November, 2008

Manuscript Submission

Dear Melissa Norton, MD

We are submitting our article entitled: “Prevalence of fibromyalgia in low socioeconomic status population”, for BMC musculoskeletal Disorders as original article. We also confirm that our manuscript has not been submitted and is not simultaneously being submitted elsewhere.

This study had Public Financial Support of: Fundação de Amparo a Pesquisa do Estado de São Paulo (FAPESP) – Foundation of Research Support of São Paulo State – and Conselho Nacional de Pesquisa e Desenvolvimento (CNPq) – National Consult of Research and Development and there is not conflict of interest to the work.

The corresponding author is: Ana Assumpçao/ Amélia Pasqual Marques – Departamento de Fisioterapia, Fonoaudiologia e Terapia Ocupacional. Rua Cipotânea, 51, Cidade Universitária, São Paulo (SP), Brazil. Zip Code: 05360-000. Phone: (+55 11) 3091-7451; Cel. (+55 11) 8120-0506; Fax. (+55 11) 3091-7462. E-mail: anassumpcao@usp.br; pasqual@usp.br

Best whishes

Ana Assumpção             Sullen Decario Chalot
Alane Bento Cavalcante    Carlos Alberto Bragança Pereira
Cristina Esteves Capela   Amélia Pasqual Marques
Juliana Ferreira Sauer